Author’s response to reviews

Title: Linc01234 promotes cell proliferation and metastasis in oral squamous cell carcinoma via miR-433/PAK4 axis

Authors:

Deyu LIU (15216250@qq.com)

xinchun jian (jian_xinchun@163.com)

PU XU (hnxupu@163.com)

Rong ZHU (702904290@qq.com)

Yuan WANG (705589852@qq.com)

Version: 4 Date: 11 Jan 2020

Author’s response to reviews:

1 - Any manuscript submitted to a BioMed Central journal must be original and the manuscript, or substantial parts of it, must not be under consideration by any other journal. We note that the current submission contains some textual overlap with other previously published works. Please re-write these sentences and phrases in your own words to ensure no overlap. If there is overlap in the Methods section, please ensure that you summarize the methods and cite the source.

We recommend the authors use an anti-plagiarism software, such as turnitin or other freely available ones, to assess any overlap and reduce it.

Response: Thanks for your kind advices! We have revised our manuscript, especially in method part.